

Temi di discussione

(Working Papers)

Issuing bonds during the Covid-19 pandemic: is there an ESG premium?

by Fabrizio Ferriani





Temi di discussione

(Working Papers)

Issuing bonds during the Covid-19 pandemic: is there an ESG premium?

by Fabrizio Ferriani

Number 1392 - November 2022

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Antonio Di Cesare, Raffaela Giordano, Monica Andini, Marco Bottone, Lorenzo Braccini, Luca Citino, Valerio Della Corte, Lucia Esposito, Danilo Liberati, Salvatore Lo Bello, Alessandro Moro, Tommaso Orlando, Claudia Pacella, Fabio Piersanti, Dario Ruzzi, Marco Savegnago, Stefania Villa. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 2281-3950 (online)

Designed by the Printing and Publishing Division of the Bank of Italy

ISSUING BONDS DURING THE COVID-19 PANDEMIC: IS THERE AN ESG PREMIUM?

by Fabrizio Ferriani*

Abstract

We rely on the ESG ratings assigned by four distinct agencies (MSCI, Refinitiv, Robeco, and Sustainalytics) to study the link between ESG scores and firms' cost of debt financing during the Covid-19 pandemic. We document the existence of a statistically and economically significant ESG premium, i.e. better rated companies access debt at a lower cost. Despite some differences across rating agencies, this result is robust to additional controls for the issuers' credit standing as well as several bond and issuer characteristics. We find that this effect is mainly driven by firms domiciled in advanced economies, whereas creditworthiness considerations prevail for firms in emerging markets. Lastly, we show that the lower cost of capital for highly rated ESG firms is explained both by investors' preference for more sustainable assets and by risk-based considerations unrelated to firms' creditworthiness, such as exposure to climate change risks.

JEL Classification: G12, G23, G32, G4.

Keywords: ESG scores, Covid-19, bond yield spreads, risk channel, non-pecuniary channel **DOI**: 10.32057/0.TD.2022.1392

Contents

1. Introduction	5
2. Literature	9
3. Data	
4. Results	
5. Channels of the ESG yield premium	
6. Conclusions	
References	

^{*} Bank of Italy, DG for Economics, Statistics and Research, Bank of Italy.

1 Introduction¹

In the early months of 2020 the spread of the Covid-19 pandemic triggered an abrupt reaction in global financial markets with pervasive and unprecedented impacts across geographical regions and asset classes. The most dramatic phase of the financial turbulence occurred between 21 February 2020, when the first lockdown measures were announced in Italy, and the last week of March, when unprecedented central banks' support measures prompted a gradual turnaround in global financial markets. During these weeks, the MSCI world market index dropped by 33%, the VIX index jumped by more than 44 points (reaching a record high value of 82), the global EMBI spread rose by around 370 basis points, the dollar appreciated by almost 3%, and portfolio outflows from funds investing in emerging markets exceeded 80\$ billions.²

Despite this period of major turmoil and acute economic uncertainty, global bond issuance was extremely abundant, see Figure 1. Notably, the first three quarters of 2020 ranked as those with the largest global debt issuance on record until then, both in terms of proceeds and number of offerings, with a steep acceleration since late March as firms sped up their refinancing process to profit from the easing of global financial conditions prompted by central banks worldwide (IMF, 2020a). Interestingly, and contrary to similar episodes of market downturn, this surge in bond issuance was not limited to bonds rated A or higher, but also extended to bond issues with higher credit risk (Halling et al., 2020a).

In this paper we focus on the bond issuance during the early stages of the Covid-19 pandemic and study whether the environmental, social and governance (ESG) profiles of firms affect their funding costs beyond what could be explained by corporate fundamentals and bond characteristics.³ This allows us to make inference on the asset-pricing implications of sustainability throughout a period of extreme turbulence as well as during the subsequent phase of market recovery. In general, ESG scores should matter as determinants of firms' funding costs to the extent that they are able to identify some components of intrinsic corporate risk (e.g. firms' exposure to climate risk) or are useful to capture investors' preferences towards more sustainable financial assets. The unexpected and exogenous nature of the shock triggered by the Covid-19 pandemic offers an ideal setting to investigate the interlink between ESG attributes and financial conditions (Albuquerque et al., 2020), and in particular to assess the resilience of sustainable assets in times of crisis. As a matter of fact, the rapid and abrupt reaction of global financial markets during the first months of

¹We thank Rui Albuquerque, Enrico Bernardini, Johnny Di Giampaolo, Marcin Kacperczyk, Filippo Natoli, Luboš Pástor, Dario Ruzzi, Giovanni Veronese, Andrea Zaghini, two anonymous referees, seminar participants at Imperial College, Sustainable macro conference, ISEFI 2022, FMCG conference 2022, Finance research letters 2022 annual conference, and Bank of Italy for useful suggestions. All remaining errors are our owns.

²See FSB (2020), Ramelli and Wagner (2020), IMF (2020b), Ding et al. (2021), Gormsen and Koijen (2020), and Ferriani (2021) among many others for some analyses of the pandemic-induced market turmoil.

³In the following we will use the terms ratings, score, ranking, profile, attributes as interchangeable.



Figure 1: Bond issuance and MSCI world dynamics

Bond issuance and MSCI world index dynamics. The graph displays the weekly amount of bond issuance between the 20th of January and the 30th of June 2020; see Section 3 for details about the bonds included in the sample. MSCI world index data come from Refinitiv and are normalized at 100 on the 20th of January 2020.

2020 limited the ability of firms to respond to the crisis, so that any relation between ESG scores and cost of funding must necessarily reflect firms' preexisting conditions, ultimately minimizing the endogeneity concerns related to the joint determination of credit spreads and ESG attributes. We study the primary bond market because it allows a more direct analysis of the corporate funding costs and is less likely to be affected by liquidity issues compared to the secondary bond market. We measure firms' funding costs in terms of asset swap spreads (ASS) and combine these data with the ESG ratings developed by four different data providers: MSCI, Refinitiv, Robeco, and Sustainalytics. With the only exclusion of MSCI, our dataset also allows to dissect the analysis not only at the aggregate level, i.e. in terms of the composite ESG score, but also with respect to the three distinct components of the ESG rating. As discussed in Section 2, previous research has found some disagreement across ESG ratings so the availability of multiple scores is essential to exclude that any interplay between ESG scores and corporate funding costs is driven by the sustainability assessment made by a particular rating agency.

Our empirical analysis highlights several results of interest. First, we find that the ESG scores

are unconditionally correlated with the ASS or, in other words, firms with better ESG attributes achieved lower financing costs during the Covid-19 crisis. When an extensive set of controls related to bond and firms' characteristics is also included in the model, then the negative and statistically significant link between ESG scores and bond risk premia somehow weakens but it is still confirmed for MSCI, Robeco, and Sustainalytics. The impact of the ESG rating, where statistically significant, is nevertheless not negligible in economic terms; for example a one standard deviation increase in the MSCI ESG score is associated with approximately 14 basis point decrease in the ASS, around 7% of the average yield spread at issuance. A similar result is also obtained when multiple ESG scores are combined via principal component analysis (PCA) in order to maximize the informative content across different rating agencies.

Second, we replicate the baseline results but we split the sample to evaluate the relation between ASS and ESG scores with respect to firms' domicile and time phases of the Covid-19 crisis. We find strong evidence of geographical segmentation in the impact of ESG scores when distinguishing observations with respect to firms' nationality. Our estimates point to a more sizable effect for firms domiciled in advanced economies (AEs) rather than in emerging market economies (EMEs), a result possibly driven by differences in investors' tastes for sustainability and in the development of the ESG industry across geographical areas. As a collateral finding, we document that issuer's creditworthiness plays a crucial role in explaining ASS variation for firms domiciled in EMEs. The coefficient of the S&P rating for bonds issued in EMEs is more than twice the corresponding estimate for firms in AEs, with a sizable economic magnitude amounting to about 110 basis points of lower ASS for a one standard-deviation increase in the issuer's credit rating. Moreover, we find evidence of a structural break in the pricing of ESG factors across the different stages of the Covid-19 crisis, with more sustainable firms achieving lower yield spreads especially during the most acute phase of market crash, a result emphasizing how sustainability enhanced corporate resilience in times of crisis.

Third, we explore the role of the individual components of the ESG rating, namely the environmental, social, and governance scores, as determinants of the bond yield spreads. As a matter of fact, the composite ESG score signals the attitude of firms towards an extensive list of "sustainable purposes" such as preserving the environment, pursuing energy efficiency, enhancing employee welfare, mitigating controversies with customers and suppliers, sustaining board independence, and strengthening governance mechanisms. To discern the specific contribution of each ESG subcomponents on firms' financing costs we regress the ASS on the three individual scores (E, S, G) assigned by each data provider. In general, our results show a very limited informativeness of the ESG subcomponents, with the partial exception of the individual E- score. Their combination via principal component analysis (PCA) to develop a synthetic composite ESG score nevertheless confirms the negative link between sustainability and bond yield spreads. Finally, in view of the general evidence pointing to a negative relation between ESG performance and corporate funding costs, we devote the last part of the paper to examine the possible channels shaping the relation between ESG scores and the ASS. Following previous theoretical contributions in this field, we examine the asset-pricing implications of two distinct channels. The first one accounts for the impact of investors' preference towards sustainability (non-pecuniary channel), whereas the second one relates to the ability of more sustainable firms to hedge against climate shocks or unexpected deteriorations in the environmental or the social dimension of corporate activity (risk-channel). To the best of our knowledge, this paper is the first to provide some empirical evidence of the interlink between these two channels and bond yield spreads. We proxy the contribution of the first factor using the share of sustainable funds holding the bond issuer's stock and the second factor using both emissions intensity and a measure of exposure to climate change extracted from the earning call transcripts and developed by Sautner et al. (2022a). We do not find any of the two channels mutually excluding the other; on the contrary, they both played a role to determine the corporate cost of funding during the Covid-19 pandemic, with an impact of around 32 basis points for the non-pecuniary channel and approximately 15 basis points for the risk channel.

Our study contributes to the literature investigating the relationship between firms' sustainability profiles and corporate cost of capital; in this regard, we deliver multiple valuable insights and implications by focusing on the primary bond-market during a period of major crisis such as the one induced by the Covid-19 pandemic. First and from the perspective of the bond issuers, we show that firms with better ESG profiles are generally able to finance their activity at a lower cost. This result is obtained by observing yield spreads throughout a period of abundant corporate bond issuance combined with elevated market uncertainty and it is robust to the inclusion of several controls, in particular a proxy of corporate credit standing. Second, from a demand perspective, the attention towards sustainable investments is part of a trend that has shaped the financial industry in recent times and has remarkably accelerated since the onset of the Covid-19 pandemic, see for instance ECB (2020), GSIA (2021), and Mohommad and Pugacheva (2022). As a result, an increasing amount of sustainable investments has been included in global portfolios (almost 36% at the end of 2020 according to GSIA, 2021) and financial service providers announced the launch of equity and bond indexes explicitly tracking companies with high ESG profiles.⁴ Even more interestingly, ESG considerations and benchmarking have been integrated in the portfolio decisions of central banks (Bernardini et al., 2021) and in their collateral framework for monetary policy operations.⁵ We document that this shift in investors' preference is not just a tendency of

⁴See for example the launch of the S&P 500 ESG index in 2019, the announcement in 2021 of two Euronext ESG indexes targeting the French and the Italian stock exchanges respectively, and the launch in 2021 of a Citi equity benchmark tracking best-in-class ESG performers across global markets.

⁵See the ECB decisions at https://bit.ly/2WrhBrW and shorturl.at/fhP06.

the industry but it has also important implications in terms of asset pricing, with more sustainable firms benefiting from lower cost of capital. Lastly, from a regulatory and policy perspective, our study also highlights some degree of heterogeneity across rating agencies in the assessment of firms' ESG profiles and this must be necessarily taken into account when analyzing the relation between ESG scores and the price of financial assets. The availability of multiple ESG scores is an infrequent feature of most of the empirical research in this field, and this makes us confident that the evidence and the conclusions presented in this paper are not flawed by some data limitations. However, the lack of a full agreement across rating providers reinforces the urgency of initiatives to overcome the current inconsistencies in the ESG classifications and deliver a common taxonomy that enhances data comparability and transparency for investors (e.g. Visco, 2019, Kleimeier and Viehs, 2021). Efforts in this regard are likely to accelerate in the near future as policy actions to embrace the ESG paradigm are spreading rapidly, see Regulation 2020/852 establishing the EU taxonomy for sustainable initiatives, the decision of the European Commission to issue around 30% of its Covid-19 recovery program (NextGenerationEU) as green bonds⁶ or the recommendation by the European Banking Authority to incorporate ESG considerations in the risk management and capital allocation process of credit institutions.⁷

The remainder of this paper is organized as follows. Section 2 briefly reviews the relevant literature for this study, Section 3 introduces the dataset and presents some descriptive statistics, Section 4 discusses the empirical results, while Section 5 focuses on the two channels affecting the corporate cost of debt. Finally, Section 6 concludes.

2 Literature

Over the last decade we have witnessed a growing body of research investigating the connection between firms' ESG attributes and multiple dimensions of corporate financial performance including but not limited to equity returns, bond yields, access to the credit market, see Friede et al. (2015), Brooks and Oikonomou (2018), Matos (2020), and Gillan et al. (2021) for some comprehensive reviews. As a matter of fact, a large part of the previous empirical studies has focused on the equity market, in particular the US one, it has limited the analysis to a single data provider of ESG ratings, and it has not explicitly investigated the link between sustainability and asset prices throughout periods of financial turbulence (a few exceptions regarding the stock market include Lins et al., 2017 and Albuquerque et al., 2020 for the global financial crisis and the Covid-19 pandemic, respectively). We concentrate on the global primary bond market during the Covid-19 crisis and we use multiple data providers to analyze the interlink between corporate sustainability

⁶See a brief summary of the program at https://bit.ly/38i6P9D.

⁷See the details of the EBA announcement at https://bit.ly/2WspFZt.

ity and debt cost of capital. The interest for this topic is at least twofold. First of all, the Covid-19 period was characterized by a drastic, although limited in time, deterioration of firms' liquidity and funding conditions, resulting in a dash-for-cash episode with direct effects on corporate risk premia and access to credit (Acharya and Steffen, 2020 and FSB, 2020). In this regard, the existence of any ESG premium for newly-issued bonds could deliver important insights for both firms and investors on the extent of corporate resilience throughout a crisis period. Second, the Covid-19 pandemic has been an incredible wake-up call for the importance of sustainability factors (e.g. Mohommad and Pugacheva, 2022) so that it is worth analyzing whether the ESG premium, where existing, has been channeled via investors' preference towards more sustainable financial assets.

In principle, two main motivations have been advanced by the literature to explain why firms chose to embrace ESG objectives and how this choice influences the price of financial assets, see Heinkel et al. (2001), Goss and Roberts (2011), Albuquerque et al. (2019), Pástor et al. (2021) Albuguerque et al. (2020), and Pedersen et al. (2021) for some contributions on the theoretical rationale of ESG investing. On one side, the interest towards firms with high ESG scores could be driven by individuals' intrinsic preference for more sustainable assets; this makes investors willing to acknowledge a premium to highly-rated ESG firms on the basis of non-pecuniary considerations. On the other side, the relation between ESG score and asset returns can be shaped by investors' conviction that ESG factors capture some source of risk that is not fully account for by traditional credit metrics, so that more sustainable assets ultimately deliver better risk-adjusted returns. On this basis, firms could be incentivized to improve their ESG profile as far as it diminishes corporate inherent risk and ensures an increase in the value of corporate assets or a reduction in income volatility. As an example, this may occur because firms with higher ESG scores are less exposed to capital and operational losses associated with a more strict environmental regulation, adverse climate events, strikes, corruption cases. As a result, ESG considerations could deliberately shape the asset allocation decisions of individuals and institutions and have important asset-pricing implications.

As concerns the interaction between ESG scores and firms' cost of debt, previous research has generally documented a lower cost of financing for more sustainable firms in terms of bond yields and credit spreads on loan data, with most of the studies explicitly targeting the US market. Bauer and Hann (2010) provide evidence that firms' credit standing and the yield spreads of newly issued bonds are influenced by the exposure of US firms to environmental concerns. Oikonomou et al. (2014) investigate the impact of several distinct dimensions of sustainability in the primary and secondary US bond market and find lower risk premia and higher credit ratings for firms with less social transgressions. Ge and Liu (2015) present similar evidence also focusing on the US primary market but using aggregate measures of corporate sustainability and adopting an IV approach to mitigate endogeneity concerns between ESG performance and bond yield spreads determination. Goss and Roberts (2011) and Chava (2014) respectively report that corporate social responsibility (CSR)⁸ and lack of environmental concerns also matter for corporate access to bank credit, with better positioned firms achieving lower spreads on their loans. Díaz and Escribano (2021) show that "green" energy firms, identified as those included in the Dow Jones Sustainability Index, benefit from a sustainability premium when issuing bonds as opposed to their "brown" counterparts. Results are partially mixed as concerns the European bond market: Menz (2010) document marginal statistical significance of the connection between CSR and the price of bonds, Stellner et al. (2015) show that sustainable firms are rewarded in terms of rating and yield spreads only to the extent that they operate in a sustainability friendly environment as expressed by the ESG profiles of their corresponding countries, while La Rosa et al. (2018) concentrate only on the social performance score and find that it reduces firms' cost of debt in normal times.

As already mentioned, being the nature of the Covid-19 shock essentially exogenous (Albuquerque et al., 2020), the crisis induced by the pandemic should represent an ideal framework to test the presence of a sustainability premium. The analysis of the connection between ESG factors and the performance of financial assets during the Covid-19 turmoil is rapidly accelerating and extending to several fields. Ferriani and Natoli (2020) and Pástor and Vorsatz (2020) show that investment funds with more sustainable assets attracted a larger amount of net inflows and delivered better returns during the Covid-19 crisis as compared to funds with high ESG risk. Ding et al. (2021) find that the drop in stock returns across 61 countries during the first five months of 2020 was milder for firms with stronger CSR activities prior to the pandemic, emphasizing the role of sustainability as a resilience factor throughout periods of heightened market stress. A similar evidence is also reported by Albuquerque et al. (2020) who expand the analysis to cover return volatility and several measures of operating performance and confirm the superior dynamics of firms with better environmental and social scores in the US equity market. Halling et al. (2020b) do not explicitly target the Covid-19 period in their analysis of the bond yields in the primary market. Contrary to us, they do not rely on ESG ratings but on a unique measure of sustainability built by aggregating firms' strengths and concerns over environmental and social attributes using company data available at the end of 2018. Most importantly, and at odds with evidence reported in this study, they do not find that firms with higher sustainability scores benefited from lower bond spreads during the pandemic crisis.

Finally, our paper also relates to the research focusing on ESG score misalignment across different data providers, see Dorfleitner et al. (2015), Berg et al. (2022), Billio et al. (2020), and Gibson et al. (2021) for some contributions on this topic. Although we do not deliberately tackle the mo-

⁸The boundaries and the differences between CSR and ESG are not immediate to outline and are nevertheless beyond the scope of this research; consistently with other contributions in the literature (e.g. Gillan et al., 2021), and without loss of generality, in this study we consider the two terms as synonyms.

tivations behind the divergence across various ESG ratings, one major advantage of this study is the availability of several ESG metrics. The simultaneous adoption of multiple ESG scores is designed to avoid that our findings and conclusions are driven by the use of a specific score. This is not a minor issue being the evidence of the interplay between firm performance and ESG attributes somewhat mixed in the literature (Gillan et al., 2021). We generally uncover a very robust evidence of a negative relation between sustainability and corporate cost of debt, but our empirical analysis nevertheless confirms the existence of some disagreement across rating agencies, which is probably more apparent when the sample is broken down with respect to firms' nationality or ESG subcomponents. Hence, our study offers some new evidence in support of academic and policy contributions calling for greater methodological transparency and comparability in the measurement of firms' ESG performance (e.g. Visco, 2019, Boffo and Patalano, 2020, Iosco, 2021, and Kleimeier and Viehs, 2021).

3 Data

We rely on multiple sources to create our dataset. Data on the global issuance of bonds by nonfinancial firms are obtained from Refinitiv, starting from the 20th of January 2020 (the date of confirmed human-to-human transmission of Covid-19) to the 30th of June 2020, covering the most acute phase of the crisis as well as the financial markets recovery of the pandemic-induced losses. For each issuer, we also use Refinitiv to retrieve a wide range of firm characteristics (e.g. total revenues, leverage, dividend yields...); all these data refer to the last available fiscal year. Then, we add information on the composite ESG score at the issuer level, which is obtained from four different data providers: MSCI, Refinitiv, Robeco, and Sustainalytics. MSCI ESG scores are collected from the MSCI website via its ESG ratings corporate search tool, Refinitiv (Asset4) scores are retrieved from Eikon, Robeco ESG scores are obtained from Bloomberg, whereas the Sustainalytics scores are collected from Morningstar. MSCI ESG scores are measured via a 7-step rating scale ranging from CCC (lowest score) to AAA (highest score); the other data providers develop a rating scheme ranging from 0 to 100, with higher values associated with superior ESG standing. For our analysis we consider the last available score at the end of 2019, i.e. the score available to investors immediately before the outburst of the Covid-19 pandemic. The coverage of bond issuers is remarkably high across the four rating agencies, ranging from around 83% in the case of MSCI to approximately 90% for Robeco and Sustainalytics, and up to almost 98% in the case of Refinitiv.9 In the case of Refinitiv, Robeco, and Sustainalytics we also have access to the three

⁹Starting from October 2019 Sustainalytics replaced its ESG sustainability ratings with company-level ESG risk scores measuring the degree to which a company's economic value may be at risk driven by materially relevant ESG factors. To ensure comparability across rating agencies, the Sustainalytics scores refer to September 2019, i.e. the last

	Ν	Mean	St.Dev.	25p	50p	75p
MSCI ESG	899	4.1	1.6	3.0	4.0	6.0
Refinitiv ESG	1052	64.9	18.1	55.6	69.7	78.0
Robeco ESG	969	52.4	29.1	29.0	54.0	76.0
Sustainalytics ESG	994	62.7	10.7	55.0	63.0	70.0
S&P issuer rating	839	10.3	2.6	9.0	10.0	12.0
Refinitiv E	1052	63.5	23.9	51.3	70.7	81.4
Refinitiv S	1052	67.8	21.6	56.1	74.2	83.3
Refinitiv G	1052	61.1	20.9	46.0	63.6	77.7
Robeco E	969	53.4	28.8	31.0	54.0	78.0
Robeco S	969	51.9	29.2	26.0	52.0	76.0
Robeco G	969	51.4	28.9	27.0	53.0	77.0
Sustainalytics E	915	65.2	13.3	55.0	66.0	76.0
Sustainalytics S	915	61.7	12.6	52.0	61.0	72.0
Sustainalytics G	915	65.4	9.5	59.0	66.0	72.0
Asset swap spread (bps)	1078	214.1	161.4	98.3	179.9	273.0
Tenor (years)	1073	9.7	8.4	5.0	7.0	10.0
Amount issued (US\$ millions)	1078	751.0	677.9	200.0	600.0	1000.0
Advanced economy	1078	0.7	0.5	0.0	1.0	1.0
Dividend yield(%)	1003	2.7	1.9	1.3	2.3	3.7
Revenues (US\$ billions)	1078	39.0	53.9	6.9	18.6	51.3
Leverage (%)	1078	0.3	0.1	0.2	0.3	0.4
Interest coverage ratio	1063	11.8	18.1	3.7	7.0	12.3
Price to book	1062	3.6	8.7	1.2	2.4	5.1
Share sust. funds (%)	1078	0.2	0.1	0.1	0.2	0.2
CO2/Total Assets	870	0.1	0.3	0.0	0.0	0.1
CO2/Total Revenues	870	0.0	0.0	0.0	0.0	0.0
CC exposure	859	0.0	0.0	0.0	0.0	0.0

Table 1: Descriptive statistics

Descriptive statistics for bond issued globally by non-financial corporations between 20th January 2020 and 30th June 2020. Variables *Share sustainable funds, CO2/Total Assets, CO2/Total Revenues ,* and *CC exposure* are used in Section 5 for the analysis of the channels of the ESG yield premium; we refer to that section for their definition and discussion.

individual scores related to the subcomponents of the ESG valuation, namely the environmental, social, and governance individual scores; the three dimensions are again assessed on a 0-100 scale with larger values indicating a better standing. Finally, we resort to Bloomberg to obtain data on the S&P long-term rating of the issuer and on the ASS of each bond on its placement day. The ASS represents our measure of firms' cost of debt and is defined as the difference between the bond yield and the yield of an asset swap contract with similar characteristics; the use of the ASS rather

available score before the adoption of the new ESG methodology.



Figure 2: ESG score distribution

Frequency distribution of the ESG composite scores. MSCI rating spans over a 7-step scale, the other scores are defined on a 0-100 scale.

than the yield spread computed by resorting to interpolated yield curves of sovereign securities is more appropriate for corporate market instruments and during periods of high volatility, see Zaghini (2016) and Zaghini (2021).¹⁰

To minimize the impact of outliers, we winsorize all variables at 1% on both tails; descriptive statistics on the sample are reported in Table 1. The sample consists of 1078 bonds for which at least one of the four ESG ratings is available; on average, the aggregate ESG score is approximately 4 in the case of MSCI which corresponds to a BBB rating of its 7-step scale, it is almost 65 for Refinitiv and around 63 for Sustainalytics, while it is slightly more than 52 for Robeco. As to the three subcomponents of the ESG scores we observe a very similar pattern, with Refinitiv and Sustainalytics providing comparable average ratings, generally above 60, while the three equivalent scores assigned by Robeco are generally lower and in the range 50-53. The average ASS is approximately equal to 214 basis points and the sample mainly includes long-term bonds, with the average maturity being almost equal to 10 years. The average S&P rating is larger than 10

¹⁰Bonds where the ASS is missing are excluded from the sample. The main results of this study are qualitatively similar when considering yield spreads with respect to a benchmark sovereign rate. The S&P long term rating of the issuer is remapped on a 17-step scale based on ratings observed in the sample, from 1 (CCC+) to 17 (AAA).

	S&P	MSCI	Ref.	Rob.	Sust.	Ref.	Ref.	Ref.	Rob.	Rob.	Rob.	Sust.	Sust.	Sust.
		ESG	ESG	ESG	ESG	Е	S	G	Е	S	G	Е	S	G
S&P	1.00													
MSCI	0.27	1.00												
Refinitiv	0.37	0.40	1.00											
Robeco	0.20	0.38	0.65	1.00										
Sustainalytics	0.23	0.58	0.60	0.63	1.00									
Ref. E	0.36	0.32	0.85	0.65	0.56	1.00								
Ref. S	0.33	0.40	0.91	0.66	0.61	0.75	1.00							
Ref. G	0.17	0.26	0.65	0.23	0.29	0.31	0.39	1.00						
Rob. E	0.20	0.38	0.64	0.95	0.62	0.66	0.66	0.20	1.00					
Rob. S	0.18	0.32	0.62	0.95	0.61	0.62	0.61	0.22	0.89	1.00				
Rob. G	0.16	0.35	0.61	0.94	0.55	0.55	0.62	0.27	0.86	0.86	1.00			
Sust. E	0.15	0.51	0.50	0.54	0.85	0.48	0.55	0.18	0.57	0.51	0.46	1.00		
Sust. S	0.18	0.46	0.50	0.49	0.88	0.49	0.50	0.20	0.48	0.52	0.40	0.56	1.00	
Sust. G	-0.02	0.50	0.45	0.47	0.79	0.34	0.44	0.31	0.46	0.47	0.43	0.56	0.62	1.00

Table 2: Correlation matrix of S&P and ESG ratings

The table displays the correlation matrix of the S&P issuer rating and the ESG scores defined both at the aggregate and the subcomponent level. MSCI ESG rating is only available at the aggregate level.

corresponding to a BBB+ once converted to the proprietary S&P codification, with approximately 90% of rated firms comprised in the investment grade category. Almost 70% of the bonds are issued in advanced economies and the average amount issued is around 750 USD millions.

The distribution of the composite ESG score is displayed in Figure 2. We observe some heterogeneity across rating providers: the distribution exhibits a quite evident negative skewness in the case of Refinitiv where the mode is equal to 74, whereas it is very concentrated for Sustainalytics with a slightly negative skewness and a mode equal to 67. On the contrary, the distribution tends to be somewhat more uniform for Robeco (mode equal to 66) and MSCI where the mode is equal to 3 (BB in the original scale).¹¹ Table 2 displays the correlation matrix of the S&P issuer rating and the ESG scores at the aggregate as well as at the subcomponent level. As expected the correlation is positive although not extremely high for several pairwise combination, a fact already emphasized in other empirical studies (e.g. Dorfleitner et al., 2015 and Berg et al., 2022). Correlations tend to be higher within each data provider (e.g. Refinitiv ESG composite rating vs. Refinitiv score at the subcomponent level) and across similar dimensions (e.g. Refinitiv E vs. Robeco E vs. Sustainalytics E). The correlation is generally quite low between the S&P and the ESG ratings, both with respect to the composite and the subcomponent scores.

Finally, Figure 3 presents the frequency distributions of bonds with respect to issuers' nationality

¹¹The distribution of the scores at the subcomponent level mimics quite closely the one displayed for the aggregate ESG ratings; graphical evidence is available upon request.



Figure 3: **Residence and industrial sector of the issuer**

The upper panel displays the distribution of the issuers' residence across geographical areas, the lower panel presents the distribution of issuers with respect to GICS sectors. Full list of countries included in each geographical area is available upon request; GICS sector "Consumer" include both Consumer Discretionary and Consumer Staples, whereas GICS sector "ICT" include both Information Technology and Communication Services.

and Global Industry Classification Standard (GICS) sectors. The upper plot shows that a substantial share of issuers is domiciled in advanced economies, especially in United States and Canada with a relative weight generally equal or larger than 50% across all rating providers. Around 20% of the bond issuers are located in European advanced economies, while the remaining observations are generally split between China and other Asian countries. The frequency distribution of the issuer's industrial sector is very similar across rating providers, with about a fourth or more of the observations belonging to the Consumer sector. Other sectors accounting for a sizable share of the observations are ICT and Industrial; a fraction approximately equal to a third of the sample is distributed across the remaining sectors, namely Energy, Health care, Materials, Real estate, and Utilities.

4 **Results**

To assess the impact of the ESG scores on firms' cost of debt we estimate pooled regression models of the following forms:

$$Spread_{j,i,t} = ESG \ score_i + Z_{j,i,t} + X_i + \gamma_t + \delta_i + \epsilon_{j,i,t}$$

$$Spread_{j,i,t} = E \ score_i + S \ score_i + G \ score_i + Z_{j,i,t} + X_i + \gamma_t + \delta_i + \epsilon_{j,i,t}$$

where *Spread* is the (log) asset swap spread of the bond *j* issued by firm *i* on its placement day *t*, *ESG score* is the ESG metrics assigned by one of the four data providers, *E score*, *S score*, and *G score* are the score of the three subcomponents of the composite ESG rating, $Z_{j,i,t}$ include a set of bond characteristics, X_i are firm-level regressors referring to the last available fiscal year before the pandemic, γ_t are time fixed effects controlling for the month of issuance, δ_i are geographical and industrial sector fixed effects, and ϵ is the error term. The first equation is meant to analyze the impact of the composite ESG rating on the ASS, whereas the second equation focuses on the impact of each individual component of the ESG score.

In our first empirical specifications, we limit the set of explanatory variables to the ESG composite rating in addition to industry, geographic, and time fixed effects to account for possible composition effects related to the Covid-19 period. Results are reported in Table 3 where we display standardized coefficients to compare the impact across different ESG metrics. For all rating providers we find evidence of a statistically significant relation between ESG scores and bond yield premia at issuance, with a higher score associated to a lower cost of funding. The effect is also sizable in economic terms: in the case of MSCI, a one standard deviation increase in the aggregate ESG score (approximately equal to a 1.5 rating upgrade on a 7 step scale) generates a 11% decline in the cost of debt at the issuance or alternatively 23 bps with respect to the average asset swap spread. The

	(1)	(2)	(3)	(4)
MSCI ESG	-0.116***			
	(-4.251)			
Refinitiv ESG		-0.063***		
		(-2.611)		
Robeco ESG			-0.068***	
			(-2.790)	
Sustain. ESG				-0.078***
				(-2.597)
Fixed-effects	Y	Y	Y	Y
Ν	899	1052	969	994
R^2	0.33	0.26	0.25	0.24

Table 3: Bond spreads and aggregate ESG scores: baseline results

The first line reports standardized coefficients defined as the marginal effect of a one standard deviation increase in each regressor on the log-ASS; t-statistics based on bootstrapped standard errors are displayed between parentheses in the second line. All models include industrial, geographic (geographical macroareas), and time (month of issuance) fixed-effects.

effect is similar across the remaining rating providers, but smaller in terms of magnitude: in this case a one standard deviation increase in the aggregate ESG score (or equivalently an upgrade by 11-29 points on a 100 scale depending on the provider) reduces the ASS in the range between 13 and 16 basis points.

The relative importance of the aggregate ESG ratings somewhat changes when we augment our baseline specifications with a wide set of controls referred to bond and corporate characteristics, and in particular with a synthetic measure of the issuer's creditworthiness, namely the S&P issuer rating. Estimates are reported in Table 4 where several results are worth emphasizing. First of all, we notice that firm's creditworthiness turns out to be the most important driver of the ASS, with a marginal effect outpacing the one of the ESG score: a one standard deviation increase in the S&P issuer rating (corresponding to about a 2.5 notch upgrade) diminishes the ASS between 65 and 77 basis points on average, depending on the specification. This is not a totally unexpected result as investors are likely to assign a larger weight to indicators of credit risk when pricing new bonds. As a second remark, we notice that the ESG score generally maintains its negative connection with firms' funding costs, however only three out of the four ESG scores preserve their statistical significance, namely MSCI, Robeco, and Sustainalytics. The marginal effect is lower but not negligible in economic terms: it ranges between 4-7% of the average ASS or equivalently between 9 and 14 basis points. These estimates confirm that a better ESG standing was rewarded in terms of lower cost of debt at issuance. This result is particularly interesting as it survives the inclusion of a set of controls, the most notably being issuer's creditworthiness; in turn this suggests that

	(1)	(2)	(3)	(4)
MSCI ESG	-0.070***			
	(-2.858)			
Refinitiv ESG		0.024		
		(0.907)		
Robeco ESG			-0.044**	
			(-2.149)	
Sustain. ESG				-0.060**
				(-2.169)
S&P	-0.362***	-0.436***	-0.449***	-0.434***
	(-15.935)	(-20.545)	(-21.145)	(-19.430)
Firm size	0.002	-0.031	-0.006	-0.030
	(0.063)	(-1.076)	(-0.248)	(-1.188)
Leverage	-0.013	-0.022	-0.041*	-0.049**
-	(-0.485)	(-0.971)	(-1.827)	(-2.165)
Dividend	0.052**	0.057***	0.070***	0.063***
	(1.965)	(2.601)	(3.546)	(2.806)
ICR	0.014	0.002	0.000	-0.006
	(0.555)	(0.092)	(0.004)	(-0.239)
PtB	0.004	-0.007	0.001	-0.009
	(0.230)	(-0.497)	(0.075)	(-0.687)
Tenor	0.194***	0.175***	0.171***	0.185***
	(11.944)	(11.062)	(10.688)	(11.847)
Amount issued	0.164^{***}	0.169***	0.150***	0.165***
	(4.763)	(5.612)	(4.648)	(5.161)
Fixed-effects	Y	Y	Y	Y
Ν	685	743	725	724
R^2	0.60	0.61	0.63	0.61

Table 4: Bond spreads and ESG scores: augmented specifications

The first line reports standardized coefficients defined as the marginal effect of a one standard deviation increase in each regressor on the log-ASS; t-statistics based on bootstrapped standard errors are displayed between parentheses in the second line. All models include industrial, geographic, and time fixed-effects. S&P is the long-term rating of the issuer, firm size is the total revenues of the issuer in log terms, leverage is the ratio between issuer's total debt and total assets, ICR is the interest coverage ratio defined as Ebit/Interest expenses, PtB is the price to book value, dividend yield is the ratio between dividend and stock price, tenor is the maturity in years, amount issued is the (log) amount of the bond issuance expressed in USD.

ESG scores have asset-pricing implications and are able to capture some previously unexplained components of the corporate cost of funding. Third, we emphasize that the negative relation between the composite ESG scores and the ASS is statistically significant even after controlling for the issuers' industrial sectors (already included in Table 3). As the ESG scores used in the analysis are industry-adjusted, a corollary of these findings is that the within-industry heterogenity

	AE	EME	AE	EME	AE	EME	AE	EME
MSCI ESG	-0.130***	0.082						
	(-6.151)	(1.135)						
Refinitiv ESG			-0.053**	0.144^{*}				
			(-2.165)	(1.815)				
Robeco ESG					-0.077***	0.036		
					(-3.702)	(0.585)		
Sustain. ESG							-0.127***	-0.015
							(-5.410)	(-0.265)
S&P	-0.267***	-0.721***	-0.383***	-0.650***	-0.360***	-0.730***	-0.352***	-0.693***
	(-12.025)	(-6.300)	(-16.898)	(-6.659)	(-16.569)	(-7.298)	(-14.863)	(-7.373)
Controls	Y	Y	Y	Y	Y	Y	Y	Y
Fixed-effects	Y	Y	Y	Y	Y	Y	Y	Y
Ν	558	127	599	144	575	150	566	158
<i>R</i> ²	0.57	0.77	0.54	0.75	0.56	0.76	0.57	0.75

Table 5: Bond spreads and ESG scores: AEs vs EMEs

The first line reports standardized coefficients defined as the marginal effect of a one standard deviation increase in each regressor on the log-ASS; t-statistics based on bootstrapped standard errors are displayed between parentheses in the second line. All models include industrial and time fixed-effects. S&P is the long-term rating of the issuer. Controls include variables to account for firms' size, leverage, ICR, price to book, dividend yield, tenor and amount of the bond issue; see Table 4 for the exact definition of each regressor.

is also relevant to explain the variation in bond yield premia. This evidence is thus alternative to Pástor et al. (2022) who find that the superior performance of green stocks is mainly driven by industry-level rather than within-industry greenness.¹² Most of the remaining controls do not display a statistically significant results as both the S&P rating and the regression fixed effects are likely to account for most of the heterogeneity in the data; we nevertheless find that firms paying larger dividends and bonds with longer maturities or larger amounts outstanding are generally associated with a higher ASS. Lastly, but not unexpectedly, we notice that the model goodness of fit achieves a remarkable improvement and increases by around three decimal points across all specifications.¹³

In Table 5 we estimate our augmented specifications but distinguish between firms domiciled in advanced and emerging market economies. To save space we only present the estimates referred to the composite ESG score and the S&P issuer rating; the full set of results is available upon

¹²We do not claim any superiority on this point but we simply want to emphasize a different finding. We clearly acknowledge that several factors could explain this contrasting result, not ultimately differences in terms of sample and time span.

¹³Estimates, available upon request, are qualitatively robust to the inclusion of additional regressors such as the currency of bond issuance.

request. Table 5 offers some important insights on the previous findings. For the subsample of firms domiciled in AEs, the impact of the ESG composite score is always negative and statistically significant; moreover, the marginal effect substantially increases, suggesting that the subsample of EMEs firms actually generates some sort of dilution effect of the results obtained on the whole set of observations. The picture for the subsample of bonds issued in EMEs turns out to be quite different: a higher ESG score is generally not supportive of a lower cost of debt. As mentioned in the Introduction, this result could not only reflect differences in terms of firm coverage and rating methodologies but it could suggest that ESG valuations in EMEs are not as common and developed as the ones referred to firms domiciled in AEs, also possibly because of the different level of firms' voluntary disclosure that is crucial for ESG assessment (Ilhan et al., 2021, van der Lugt et al., 2020). Alternatively, and under the assumption of home-investor bias, firms issuing in AEs are more likely to face investors with broader and more complex preferences embracing social responsibility beside standard risk-return considerations; this could also explain the premium in terms of lower ASS for more sustainable firms domiciled in AEs. As a second remark, it is worth emphasizing the different impact of issuer's creditworthiness across the two subsamples. The S&P coefficient for bonds issued in EMEs is larger than the corresponding estimate for AEs by a factor ranging between 1.7-2.7; this amounts to up to around 111 basis points of lower ASS for a one standard deviation increase in the S&P rating. In other words, geographical differentiation also matters in terms of the relative weight assigned by investors to firms' credit standing, with a larger penalization for firms domiciled in countries where financial conditions are possibly more fragile, financial markets are generally less developed, and corporate disclosure less established. In Table 6 we study the impact of ESG scores on bonds' ASS across different stages of the Covid-19 pandemic and to this purpose we split the sample into two distinct sub-periods. Consistently with other empirical research on the financial impacts of Covid-19 (Ramelli and Wagner, 2020), the first period includes bonds issued from 21 February 2020, when the first lockdown measures were announced in Italy, to the end of March 2020, when the Federal Reserve announced major interventions in the corporate bond market. This period, labeled as crisis in the table, covers the most intense phase of the stock market crash induced by the Covid-19 pandemic.¹⁴ The second period, referred to as the recovery phase, encompasses bonds issued from April to June 2020 and it embraces the stage of gradual rebound in global financial markets. Table 6 shows that the ESG attributes are more priced in bond yields during the most acute phase of the crisis, whereas no statistically significant effect is generally found during the recovery phase with the exception of MSCI. Interestingly, and contrary to what is observed for the ESG score, the coefficient of the S&P rating is statistical significant across rating providers and market stages, with its magnitude be-

¹⁴Results, available upon request, are qualitatively similar if the crisis sub-period includes also bonds issued between the 20 of January (start of the sample) and the 21 of February.

	Crisis	Recov.	Crisis	Recov.	Crisis	Recov.	Crisis	Recov.
MSCI ESG	-0.119**	-0.076**						
	(-2.278)	(-2.478)						
Refinitiv ESG			-0.125***	0.049				
			(-2.710)	(1.459)				
Robeco ESG					-0.057	-0.018		
					(-1.299)	(-0.787)		
Sustain. ESG							-0.136***	-0.030
							(-2.821)	(-0.988)
S&P	-0.178***	-0.413***	-0.246***	-0.479***	-0.275***	-0.485***	-0.289***	-0.477***
	(-4.054)	(-16.323)	(-5.161)	(-18.883)	(-6.089)	(-19.417)	(-6.180)	(-18.858)
Controls	Y	Y	Y	Y	Y	Y	Y	Y
Fixed-effects	Y	Y	Y	Y	Y	Y	Y	Y
Ν	185	429	196	464	196	444	195	445
<i>R</i> ²	0.65	0.59	0.65	0.60	0.65	0.62	0.65	0.61

Table 6: Bond spreads and ESG scores: crisis vs recovery

The first line reports standardized coefficients defined as the marginal effect of a one standard deviation increase in each regressor on the log-ASS; t-statistics based on bootstrapped standard errors are displayed between parentheses in the second line. All models include industrial, geographic, and time fixed-effects. S&P is the long-term rating of the issuer. Controls include variables to account for firms' size, leverage, ICR, price to book, dividend yield, tenor and amount of the bond issue; see Table 4 for the exact definition of each regressor. Column "crisis" presents the estimates for bonds issued from 21 February to 31 March 2020, while column "recovery" refers to bonds issued from 1 April until 30 June 2020.

ing always larger during the phase of market recovery. This finding can be possibly attributed to the steep acceleration in bond issuance started at the end of March, as displayed in Figure 1. Following a phase of pervasive market crash hitting indistinctly all categories of financial assets, investors increasingly relied on (more established) external credit assessment to discriminate among firms during a period of abundant bond issuance. Evidence of a structural break in the importance of the ESG factor is not clear-cut in the literature, and the empirical analysis is still preliminary in this regard. As concerns the equity market, Albuquerque et al. (2020) show that environmental and sustainable ratings have grown in importance to explain the differences in cumulative stock returns following the Covid-19 shock, with a sizable increase of their relative loading since late February 2020 before flattening at the end of March. Focusing on the first semester of 2020, Ferriani and Natoli (2020) find that the ESG exposure of mutual funds has been increasingly considered by fund investors during the recovery of financial markets, while Pástor and Vorsatz (2020) show that the Covid-19 pandemic did not result in a structural break of investor sustainability preferences before and after the Covid-19 crisis.¹⁵

¹⁵As a further analysis we investigate the relation between corporate cost of funding and ESG scores across dif-

	Full s.					
MSCI ESG	-0.078***					
	(-2.623)					
Refinitiv ESG	0.013					
	(0.435)					
Robeco ESG	-0.035					
	(-1.424)					
Sustain. ESG	0.008					
	(0.226)					
PCA all		-0.063**				
		(-2.377)				
PCA ex MSCI			-0.036			
			(-1.471)			
PCA ex Refinitiv				-0.073***		
				(-2.732)		
PCA ex Robeco					-0.056*	
					(-1.959)	
PCA ex Sust.						-0.060**
						(-2.443)
S&P	-0.387***	-0.395***	-0.444***	-0.393***	-0.373***	-0.392***
	(-15.587)	(-16.571)	(-19.846)	(-16.393)	(-15.364)	(-17.586)
Controls	Y	Y	Y	Y	Y	Y
Fixed-effects	Y	Y	Y	Y	Y	Y
Ν	641	641	683	641	658	664
R^2	0.63	0.62	0.64	0.62	0.60	0.63

Table 7: Bond spreads and ESG scores: joint inclusion of ESG ratings

The first line reports standardized coefficients defined as the marginal effect of a one standard deviation increase in each regressor on the log-ASS; t-statistics based on bootstrapped standard errors are displayed between parentheses in the second line. All models include industrial, geographic, and time fixed-effects. S&P is the long-term rating of the issuer. Controls include variables to account for firms' size, leverage, ICR, price to book, dividend yield, tenor and amount of the bond issue; see Table 4 for the exact definition of each regressor."PCA all" measures the first principal component from the four composite ESG scores; analogously, "PCA ex MSCI", "PCA ex Refinitiv", "PCA ex Robeco", and "PCA ex Sustainalytics" measure the first principal component from composite ESG scores but excluding one agency at a time.

Table 7 simultaneously considers the informative content of the four ESG ratings, which is substantially equivalent to estimate the model on the common set of observations across the four rating providers. To this purpose we first include all ESG ratings in the same specification. Then, and in the same spirit, we also create a synthetic ESG index by extracting the first principal component from composite ESG scores. We consider two possible alternatives for this second approach: one combining all the available scores and one where we exclude one rating agency at a time.

ferent industrial sectors, but we do not find any clear-cut evidence of sectoral-specific impacts. These results could be related to the limited number of observations available for some industries and to the fact that the across-industry analysis is likely to exacerbate the divergence across ESG ratings because of factors including, but not limited to, firm coverage and methodological assumptions such as industry-specific adjustments. Results are available upon request.

The first column of Table 7 displays a negative and statistically significant only for MSCI, with an economic magnitude non dissimilar from the one reported in Table 4. This result must not necessarily be interpreted as a ranking across different ESG metrics and we are not claiming the superiority of a specific rating provider compared to the others, although some previous research pointed out substantial differences across rating agencies in terms of firm coverage and rating use by professional investors (e.g. Eccles and Stroehle, 2018, Hirai and Brady, 2021). As a matter of fact, the estimates in Table 7 are not necessarily explained by the better informative content of one particular score but could be related to other motivations including, but not limited to, the firm coverage, the specific sample analyzed in this study, the public availability of ratings. In this regard, the use of PCA is particularly useful as it allows to exploit data variation in our sample without imposing any ex-ante ranking across rating agencies. Rather, in the assessment of the relation between ESG performance and bond yield premia, the PCA allows to maximize the informative content conveyed by different ESG scores. Depending on the specific version of the PCA index, the first principal component explains between 64% and 74% of the variance and is therefore able to capture a substantial amount of heterogeneity in the data. Columns 2-6 generally confirm the negative relation between ESG score and corporate cost of funding, albeit the synthetic proxy based on the exclusion of the MSCI score is not statistically significant.

In our last empirical exercise we investigate the relation between the ASS and the ESG rating at the subcomponent level, namely the environmental, social, and governance scores.¹⁶ To this purpose, we first re-estimate the model but we substitute the composite ESG rating with all the three individual scores (E, S, G) obtained from each data provider. Then, we only focus on the E- score, as environmental concerns are widely considered to be the ones playing the leading role across different sustainability dimensions, see for example Gibson Brandon et al. (2021), Pástor et al. (2022), Krueger et al. (2020a). Finally, we also include a synthetic composite ESG score defined as the first principal component of the E-, S-, and G- individual scores. When all the three individual scores are simultaneously included in our specification, the estimates reported in Table 8 highlight a very limited informativeness of ESG subcomponents on the corporate cost of debt, being regression coefficients not statistically significant in most of the cases. Even when statistically significant results emerge, the estimates should be interpreted with extreme caution in view of the very high level of correlation across ESG subcomponents within each rating agency; indeed, this is particularly true for Robeco ratings (correlation >0.85) as displayed by the correlation matrix in Table 2 and confirmed by a post-estimation VIF test. When we limit the analysis to the environmental score, also in order to control for possible collinearity issues, we find that greener firms generally paid lower yields spreads, at least for two out of three rating providers; the magnitude of E-score coefficient is comparable, if not even larger, to the one of the composite ESG scores displayed in

¹⁶We recall that this classification is not available for MSCI in our dataset.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Refinitiv E	-0.007	0.017							
	(-0.273)	(0.818)							
Refinitiv S	0.045								
	(1.461)								
Refinitiv G	-0.016								
	(-0.809)								
PCA Ref.			0.013						
			(0.512)						
Robeco E				-0.024	-0.049**				
				(-0.672)	(-2.216)				
Robeco S				0.156***					
				(3.392)					
Robeco G				-0.192***					
				(-4.500)					
PCA Rob.						-0.047**			
						(-2.120)			
Sustain. E							-0.139***	-0.099***	
							(-5.063)	(-4.294)	
Sustain. S							0.060**		
							(1.987)		
Sustain. G							0.042		
							(1.572)		0.011
PCA Sust.									-0.061**
	o 10 0***		0 1 - 0 * * *	0.440***		0 11 4 4 4 4	2 2 2 2 2 4 4 4 4	~ ~ ~~***	(-2.515)
S&P	-0.438***	-0.437***	-0.452***	-0.468***	-0.450***	-0.446***	-0.390***	-0.398***	-0.424***
	(-20.142)	(-20.185)	(-19.146)	(-21.941)	(-21.242)	(-18.399)	(-16.035)	(-17.462)	(-18.183)
Controls	Y	Y	Y	Y	Y	Y	Y	Ŷ	Y
Fixed-eff.	Y TAO	Y TAO	Y T10	Y	Y	Y	Y	Ŷ	Y
N P ²	743	743	743	725	725	725	694	694	694
K *	0.61	0.61	0.54	0.65	0.63	0.55	0.62	0.62	0.55

Table 8: Bond spreads and ESG subcomponents scores

The first line reports standardized coefficients defined as the marginal effect of a one standard deviation increase in each regressor on the log-ASS; t-statistics based on bootstrapped standard errors are displayed between parentheses in the second line. All models include industrial, geographic, and time fixed-effects. S&P is the long-term rating of the issuer. Controls include variables to account for firms' size, leverage, ICR, price to book, dividend yield, tenor and amount of the bond issue; see Table 4 for the exact definition of each regressor. "PCA Refinitiv", "PCA Robeco', "PCA Sustainalytics" measure the first principal component of the E-, S-, and G- subcomponent scores of each rating agency.

Table 4. On the contrary, if we replicate this exercise with the other individual scores, i.e. the Sand the G- ratings, the estimates are mostly not significant.¹⁷ All in all, these results are in line with previous empirical studies showing that individual ESG dimensions *per se* are generally not

¹⁷The only statistically significant coefficient is the Robeco G-score, which displays a negative sign. The full set of estimates is available upon request.

informative to explain the superior performance of more sustainable financial assets, at most with the exception of firms' environmental attributes which seem to catalyze most of investors' inclination towards the overall ESG paradigm, see Oikonomou et al. (2014), Albuquerque et al. (2020), Halling et al. (2020a), Pástor and Vorsatz (2020), Ferriani and Natoli (2020). Finally, we notice that the PCA index obtained from the three individual scores is able to convey an information content very close to the original composite counterpart, i.e. more sustainable firms benefit from lower cost of capital, with an impact that is also statistically significant and economically sizable for Robeco and Sustainalytics.

5 Channels of the ESG yield premium

In the previous Section, we provide a robust evidence of the negative link between ESG scores and corporate cost of funding. Despite some differences in terms of economical and statistical significance across rating agencies and subsamples, we generally find that better ESG scores are associated with lower ASS, even after controlling for an extensive list of bond and corporate characteristics, including firms' credit rating.

In this Section we extend our investigation to the channels that are likely to inform the relation between ESG scores and bond yields. Our analysis is inspired by previous contributions that outline the rationale behind the asset pricing of the ESG factor, see for example Heinkel et al. (2001), Albuquerque et al. (2019), Pástor et al. (2021), Pedersen et al. (2021). Although each study has its own peculiarities in terms of the overall theoretical framework, the literature is generally consistent in identifying two main drivers of ESG investing. On one side, investors with strong ESG preferences derive utility from holdings assets of more sustainable firms and are willing to pay a premium to include these assets in their portfolio, ultimately lowering firms' cost of capital; we refer to this as the *non-pecuniary channel*. On the other side, firms with better ESG scores are also more likely to offer a hedge against events such as climate shocks or unexpected deteriorations in the environmental regulation, so that investors require a higher compensation to hold the assets of less sustainable firms; we refer to this driver as the *risk channel*. The fact that sustainability preferences and ESG risk-related considerations are not directly observed partially explains why most of the previous empirical research has analyzed asset-pricing implications by using some comprehensive measures such as the ESG scores. We try to fill this gap by separately estimating the impact of the non-pecuniary and the risk channel on the ASS. As we recognize that some overlapping could in principle exist between these two channels, we underline that the aim of this Section is not to disentangle investors' attitude towards sustainability into two orthogonal components. Rather, and in the same spirit of Pedersen et al. (2021), we try to empirically investigate the theoretical implications of the ESG literature by using some proxies that should reasonably reflect the two channels at play.

As concerns the non-pecuniary channel, we rely on Morningstar to retrieve the share of sustainable funds holding the stock of the bond issuer in the quarter immediately before the bond issuance. We consider as sustainable those funds that are assigned a Morningstar's Sustainability rating (the so-called "globes") equal to 4 or 5 on a 1-5 scale, with previous research showing that the globe rating is successful in capturing investors' non-pecuniary motivations (e.g. Hartzmark and Sussman, 2019).¹⁸ We consider firms' equity rather than the corresponding bond issued during the Covid-19 period as the former is a much more liquid corporate asset; moreover, newly-issued bonds are likely to be included into mutual funds' portfolios with some delay and this could have some limitations in terms of exogeneity and sample representativeness. Previous literature has provided robust evidence that more sustainable funds are less characterized by flow volatility compared to conventional funds and that non-financial considerations, rather than the past financial performance, act as the main driver of investors' flows towards this type of investment vehicles (e.g. Bollen, 2007, Renneboog et al., 2011, Riedl and Smeets, 2017, Bauer et al., 2021, Bialkowski and Starks, 2016). Hence, our proxy should reveal those assets where investors tend to embody their non-pecuniary preferences.

We rely on two different approaches to proxy the risk-channel of ESG investing; both are based on corporate exposure to climate change risks. Indeed, as already discussed in the previous sections, environmental concerns are likely to play a leading role in shaping the interlink between sustainability and corporate asset prices. The first proxy is based on carbon dioxide emissions (CO₂): more polluting firms may be exposed to interventions curbing corporate emissions via regulatory acts or via carbon pricing mechanisms such as emissions trading systems and carbon taxes. Moreover, highly polluting firms could also be indicative of more conventional business model whose profitability could be impacted by the ongoing ecological transition. Bolton and Kacperczyk (2021) document that investors require a compensation for their exposure to more polluting activities, while Ilhan et al. (2020) show that the cost of protection against adverse shift in climate regulation is larger for firms with more carbon-intense business models. We base our analysis on corporate emissions intensity measured n terms of tons of Scope 1 emissions (CO₂ equivalents) over firms total assets or firm revenues.¹⁹ The second proxy for the risk-channel

¹⁸The share is computed with respect to all mutual funds holding the issuer's stock; mutual funds holding the stock but not assigned a Sustainability score are excluded from the computation.

¹⁹CO₂ emissions can be broken down into three categories: scope 1 emissions are defined as those caused directly by the organization's activities, scope 2 emissions count indirect emissions resulting from the organization's energy consumption, while scope 3 emissions are defined as all other indirect emissions that are a result of the organization's operations but are generated from sources that are not owned or controlled by the organization itself. We focus on scope 1 emissions as they are more directly linked to firm's business and are less exposed to imputation and estimations issues. Data are obtained from Carbon4Finance and refer to the latest available calendar year before 2020.

	(1)	(2)	(3)	(4)
Share sust. funds (%)	-0.146***	-0.146***	-0.160***	-0.157***
	(-3.457)	(-3.482)	(-3.877)	(-3.695)
C02/Total Assets	0.068**			
	(1.967)			
C02/Total Revenues		0.058^{*}		
		(1.759)		
CC expos.			0.067***	
			(2.918)	
CC regul. expos.				0.049**
				(1.966)
CC phys. expos.				-0.018
				(-0.254)
S&P	-0.361***	-0.367***	-0.355***	-0.364***
	(-13.177)	(-12.942)	(-14.140)	(-14.281)
Controls	Y	Y	Y	Y
Fixed-effects	Y	Y	Y	Y
N	642	642	670	670
R^2	0.64	0.64	0.63	0.63

Table 9: Bond spreads and ESG scores: underlying channels

The first line reports standardized coefficients defined as the marginal effect of a one standard deviation increase in each regressors on the log-ASS; t-statistics based on bootstrapped standard errors are displayed between parentheses in the second line. All models include industrial, geographic, and time fixed-effects. S&P is the long-term rating of the issuer. Controls include variables to account for firms' size, leverage, ICR, price to book, dividend yield, tenor and amount of the bond issue; see Table 4 for the exact definition of each regressor. *Share of sustainable funds* is the share of sustainable funds (Morningstar's Sustainability equal to 4 or 5) holding the bond issuer's equity share in the quarter immediately before the bond issuance. CO2/Total Assets and CO2/Total Revenues measure scope 1 emissions over firms' assets and revenues. CC exposure, CC regulatory exposure, CC physical exposure, and CC risk measure firm-level exposure to climate change and its corresponding degree of uncertainty, see Sautner et al. (2022a) for the exact definition of each variable.

takes advantage of the measure of climate change exposure recently developed by Sautner et al. (2022a). By applying machine learning techniques to earning calls transcripts, the authors are able to quantify *firm-level* exposure to climate change shocks. In a few words, the exposure to climate change is measured by counting the frequency with which certain climate change bigrams (i.e. pre-specified combination of words) occur in the transcript, scaled by the total number of bigrams in the transcript; we nevertheless refer to the original paper for a more precise description of the methodology. A recent study on the asset-pricing implications of this measure shows that firm-level exposure to climate change is positively correlated with equity risk premium (Sautner et al., 2022b). In the following we use the measure of climate change exposure referring to the

quarter immediately preceding the issuance of each bond; we also complement the analysis with further measures disentangling the overall exposure to climate change into its subcomponents related to regulatory and physical shocks.

To assess the impact of the two channels of ESG investing, we replace our comprehensive measures of sustainability (the ESG scores) with the various proxies described in this section;²⁰ estimates are reported in Table 9. As a first remark, we note that both drivers play a role in the determination of the ASS. The direction of the effect is clearly opposite: on one side firms potentially more exposed to climate change pay a premium on their bond yields, on the other side the premium is lower for those issuers that are able to attract the demand of investors with strong ESG appetite. Also the magnitude of the effect is not totally comparable and is larger for the nonpecuniary dimension: a one standard deviation increase in the share of sustainable funds holding the equity share of the bond issuer generates a reduction in the ASS up to 32 basis points, whereas the analogous estimate for the risk channel ranges between 12 and 15 basis points. Interestingly, the alternative approaches used to proxy the risk channel delivers very similar point estimates, despite the two measures being only mildly correlated (0.40 for CC exposure and CO_2 /Total Assets and 0.54 for CC exposure and CO_2 /Total Revenues). Although tempting, it would be pretentious to draw some general conclusions on the dominance of one channel over the other without a lengthy time span. Indeed, as shown in Bolton and Kacperczyk (2021) and Ilhan et al. (2020), the pricing of the carbon factor is not stable and evolves over time; however, our analysis focuses on a period characterized by a substantial surge in media and investors' attention towards sustainability triggered by the Covid-19 pandemic (Mohommad and Pugacheva, 2022) and this could ultimately put more weight on the non-pecuniary channel of ESG investing. Lastly, we notice that most of the impact of risk channel seems to be driven by the exposure to adverse regulatory shocks such as an unexpected tightening of the environmental regulation rather than a physical threat in line with the evidence in Krueger et al. (2020b) and Seltzer et al. (2021).

Notwithstanding the inclusion of industry fixed effects in the empirical specification, a possible critique to the results displayed in Table 9 is that our sample also consists of bonds issued by firms in highly polluting sectors. These financial assets could be assimilated to the so called "sin stocks" (Hong and Kacperczyk, 2009) and investors could be tempted to operate a preemptive exclusionary strategy on the basis of ethical considerations. If this is the case, then a proper assessment of the impact of two channels of ESG-investing could be altered. In line with the approach of Bolton and Kacperczyk (2021), Table 10 presents the estimates where we exclude the bonds issued by firms in the oil and gas, utilities, and transportation sectors (GICS Industry 2, 65-69, 19, 20 and 23). The coefficients are very similar to the ones based on the whole set of observations suggesting

²⁰The proxies for the two channels are mildly correlated (in the range of -0.22 to 0.04 depending on the variable), which is somewhat supportive of no critical overlapping in the assessment of the two channels.

	(1)	(2)	(3)	(4)
Share sust. funds (%)	-0.107**	-0.111**	-0.179***	-0.176***
	(-2.141)	(-2.231)	(-3.665)	(-3.469)
CO2/Total Assets	0.075**			
	(2.005)			
CO2/Total Revenues	. ,	0.066*		
		(1.792)		
CC expos.		`	0.048^{***}	
1			(2.680)	
CC regul. expos.				0.005
				(0.175)
CC phys. expos				-0.015
ee prije. expesi				(-0.160)
S&P	-0 375***	-0 378***	-0 342***	-0 339***
blai	(-10.782)	(-10.931)	(-11, 262)	(-10.892)
Controls	(10.702) V	(10.551) V	(11.202) V	(10.0)2)
Controls	I	I	1	I
Fixed-effects	Y	Y	Y	Y
Ν	524	524	543	543
R^2	0.61	0.61	0.63	0.63

Table 10: Bond spreads and ESG scores: underlying channels - no polluting sectors

The first line reports standardized coefficients defined as the marginal effect of a one standard deviation increase in each regressor on the log-ASS; t-statistics based on robust standard errors are displayed between parentheses in the second line. All models include industrial, geographic, and time fixed-effects. S&P is the long-term rating of the issuer. Controls include variables to account for firms' size, leverage, ICR, price to book, dividend yield, tenor and amount of the bond issue, and corporate carbon intensity; see Table 4 for the exact definition of each regressor. *Share of sustainable funds* is the share of sustainable funds (Morningstar's Sustainability equal to 4 or 5) holding the bond issuer's equity share in the quarter immediately before the bond issuance. CO2/Total Assets and CO2/Total Revenues measure scope 1 emissions over firms' assets and revenues. CC exposure, CC regulatory exposure, CC physical exposure, and CC risk measure firm-level exposure to climate change and its corresponding degree of uncertainty, see Sautner et al. (2022a) for the exact definition of each variable.

that the impact of the risk channel on the ESG premium is not driven by a divestment effect from more polluting sectors. The unique difference concerns the coefficient of the variable capturing the impact of a regulatory shock linked to climate change which is now insignificant. In turn, this could suggest that investors consider the overall assessment of ESG related risks as a relevant factor to be priced in, while a more narrow proxy explicitly focusing on the adverse shifts in climate regulation is material only for more polluting industries.

6 Conclusions

We study the relation between corporate ESG scores and the yield spreads of bonds issued by global non-financial corporations during the first phase of the Covid-19 crisis. Despite the episode of major turbulence in international financial markets, corporate activity in the primary bond market was substantial as firms strove to alleviate the funding and liquidity strains induced by the pandemic. Our analysis investigates the presence of a sustainability premium to be intended as a lower cost of debt for firms meeting ESG objectives. To this purpose we rely on ESG scores assigned by four distinct rating agencies (MSCI, Definitive, Robeco, and Sustainalytics) and highlight several new results of interest. First, we generally document a negative relation between aggregate ESG scores and debt cost of funding; even when a measure of firm's creditworthiness is explicitly included in the empirical specifications we find that the connection between ESG profiles and bond risk premia is economic meaningful and amounts to up to 14 basis points or approximately 7% of the sample average asset swap spread. Second, we find that the ESG premium is remarkably more sizable for firms domiciled in advanced economies whereas the issuer's creditworthiness is largely regarded as the most critical determinant of corporate cost of capital in emerging market economies. We show that the ESG factor was priced in bond yields especially during the phase of the market crash rather than the recovery stage; moreover, we find that the ESG individual subcomponents *per se* have generally little power to explain the heterogeneity in bond yield spreads, with the partial exception of the E- score. Finally, but not less importantly, we empirically test the theoretical rationale behind the ESG premium and show that both non-pecuniary motivations as well as risk-based considerations explain investors' preference towards more sustainable assets. The findings reported in this paper set the stage for future research avenues including, but not limited to, the impact of the two channels of ESG investing on financial instruments beyond corporate bonds or the analysis of the role played by new categories of investors (e.g. central banks) in the ESG arena. In terms of policy implications, our study underscores how the divergence across ESG ratings should be carefully taken into consideration when studying the impact of ESG scores on the price of financial assets, ultimately supporting the initiatives to rapidly achieve a common and transparent taxonomy of the ESG attributes.

References

- Acharya, V. V. and Steffen, S. (2020), "The risk of being a fallen angel and the corporate dash for cash in the midst of COVID," *The Review of Corporate Finance Studies*, 9, 430–471.
- Albuquerque, R., Koskinen, Y., and Zhang, C. (2019), "Corporate social responsibility and firm risk: Theory and empirical evidence," *Management Science*, 65, 4451–4469.
- Albuquerque, R., Koskinen, Y., Yang, S., and Zhang, C. (2020), "Resiliency of environmental and social stocks: An analysis of the exogenous COVID-19 market crash," *The Review of Corporate Finance Studies*, 9, 593–621.
- Bauer, R. and Hann, D. (2010), "Corporate environmental management and credit risk," SSRN working paper.
- Bauer, R., Ruof, T., and Smeets, P. (2021), "Get real! Individuals prefer more sustainable investments," *The Review of Financial Studies*, 34, 3976–4043.
- Berg, F., Kölbel, J. F., and Rigobon, R. (2022), "Aggregate Confusion: The Divergence of ESG Rating," *Review of Finance*.
- Bernardini, E., Faiella, I., Lavecchia, L., Mistretta, A., and Natoli, F. (2021), "Central banks, climate risks and sustainable finance," *Bank of Italy Occasional paper*.
- Bialkowski, J. and Starks, L. T. (2016), "SRI funds: Investor demand, exogenous shocks and ESG profiles," *working paper, University of Texas at Austin*.
- Billio, M., Costola, M., Hristova, I., Latino, C., and Pelizzon, L. (2020), "Inside the ESG Ratings:(Dis) agreement and performance," *University Ca'Foscari of Venice, Dept. of Economics Research Paper Series No*, 17.
- Boffo, R. and Patalano, R. (2020), "Esg investing: Practices, progress and challenges," *Éditions* OCDE, Paris.
- Bollen, N. P. (2007), "Mutual fund attributes and investor behavior," *Journal of financial and quantitative analysis*, 42, 683–708.
- Bolton, P. and Kacperczyk, M. (2021), "Do investors care about carbon risk?" *Journal of Financial Economics*.
- Brooks, C. and Oikonomou, I. (2018), "The effects of environmental, social and governance disclosures and performance on firm value: A review of the literature in accounting and finance," *The British Accounting Review*, 50, 1–15.

- Chava, S. (2014), "Environmental externalities and cost of capital," *Management science*, 60, 2223–2247.
- Díaz, A. and Escribano, A. (2021), "Sustainability premium in energy bonds," *Energy Economics*, 95, 105113.
- Ding, W., Levine, R., Lin, C., and Xie, W. (2021), "Corporate immunity to the COVID-19 pandemic," *Journal of Financial Economics*, 141, 802–830.
- Dorfleitner, G., Halbritter, G., and Nguyen, M. (2015), "Measuring the level and risk of corporate responsibility–An empirical comparison of different ESG rating approaches," *Journal of Asset Management*, 16, 450–466.
- ECB (2020), "Financial Stability Review," November.
- Eccles, R. G. and Stroehle, J. C. (2018), "Exploring social origins in the construction of ESG measures," *University of Oxford working paper*.
- Ferriani, F. (2021), "From taper tantrum to Covid-19: Portfolio flows to emerging markets in periods of stress," *Journal of International Financial Markets, Institutions and Money*, 74, 101391.
- Ferriani, F. and Natoli, F. (2020), "ESG risks in times of Covid-19," Applied Economics Letters, 0, 1–5.
- Friede, G., Busch, T., and Bassen, A. (2015), "ESG and financial performance: aggregated evidence from more than 2000 empirical studies," *Journal of Sustainable Finance & Investment*, 5, 210–233.
- FSB (2020), "Holistic review of the March market turmoil," FSB report.
- Ge, W. and Liu, M. (2015), "Corporate social responsibility and the cost of corporate bonds," *Journal of Accounting and Public Policy*, 34, 597–624.
- Gibson, R., Krueger, P., and Schmidt, P. S. (2021), "ESG rating disagreement and stock returns," *Financial Analysts Journal*, 77, 104–127.
- Gibson Brandon, R., Krueger, P., and Schmidt, P. S. (2021), "ESG rating disagreement and stock returns," *Financial Analysts Journal*, 77, 104–127.
- Gillan, S. L., Koch, A., and Starks, L. T. (2021), "Firms and social responsibility: A review of ESG and CSR research in corporate finance," *Journal of Corporate Finance*, 66, 101889.
- Gormsen, N. J. and Koijen, R. S. (2020), "Coronavirus: Impact on stock prices and growth expectations," *The Review of Asset Pricing Studies*, 10, 574–597.

- Goss, A. and Roberts, G. S. (2011), "The impact of corporate social responsibility on the cost of bank loans," *Journal of Banking & Finance*, 35, 1794–1810.
- GSIA (2021), "Global sustainable investment review," Global Sustainable Investment Alliance review.
- Halling, M., Yu, J., and Zechner, J. (2020a), "How did COVID-19 affect firms' access to public capital markets?" *The Review of Corporate Finance Studies*, 9, 501–533.
- Halling, M., Yu, J., and Zechner, J. (2020b), "Primary Corporate Bond Markets and Social Responsibility," *Working paper available at SSRN*.
- Hartzmark, S. M. and Sussman, A. B. (2019), "Do investors value sustainability? A natural experiment examining ranking and fund flows," *The Journal of Finance*, 74, 2789–2837.
- Heinkel, R., Kraus, A., and Zechner, J. (2001), "The effect of green investment on corporate behavior," *Journal of financial and quantitative analysis*, 36, 431–449.
- Hirai, A. and Brady, A. (2021), "Managing ESG Data and Rating Risk," *Harvard Law School Forum on Corporate Governance*.
- Hong, H. and Kacperczyk, M. (2009), "The price of sin: The effects of social norms on markets," *Journal of financial economics*, 93, 15–36.
- Ilhan, E., Sautner, Z., and Vilkov, G. (2020), "Carbon Tail Risk," *The Review of Financial Studies*, 34, 1540–1571.
- Ilhan, E., Krueger, P., Sautner, Z., and Starks, L. T. (2021), "Climate risk disclosure and institutional investors," *Swiss Finance Institute Research Paper*.
- IMF (2020a), "Bridge to recovery," Global Financial Stability Report, October.
- IMF (2020b), "Markets in the Time of COVID-19," Global Financial Stability Report, April.
- Iosco (2021), "Environmental, Social and Governance (ESG) Ratings and Data Products Providers," *Iosco report*.
- Kleimeier, S. and Viehs, M. (2021), "Pricing carbon risk: Investor preferences or risk mitigation?" *Economics Letters*, 205, 109936.
- Krueger, P., Sautner, Z., and Starks, L. T. (2020a), "The importance of climate risks for institutional investors," *The Review of Financial Studies*, 33, 1067–1111.

- Krueger, P., Sautner, Z., and Starks, L. T. (2020b), "The importance of climate risks for institutional investors," *The Review of Financial Studies*, 33, 1067–1111.
- La Rosa, F., Liberatore, G., Mazzi, F., and Terzani, S. (2018), "The impact of corporate social performance on the cost of debt and access to debt financing for listed European non-financial firms," *European Management Journal*, 36, 519–529.
- Lins, K. V., Servaes, H., and Tamayo, A. (2017), "Social capital, trust, and firm performance: The value of corporate social responsibility during the financial crisis," *the Journal of Finance*, 72, 1785–1824.
- Matos, P. (2020), "ESG and responsible institutional investing around the world: A critical review," *CFA Institute Research Foundation*.
- Menz, K.-M. (2010), "Corporate social responsibility: Is it rewarded by the corporate bond market? A critical note," *Journal of Business Ethics*, 96, 117–134.
- Mohommad, A. and Pugacheva, E. (2022), "Impact of COVID-19 on attitudes to climate change and support for climate policies," *IMF Working paper*.
- Oikonomou, I., Brooks, C., and Pavelin, S. (2014), "The effects of corporate social performance on the cost of corporate debt and credit ratings," *Financial Review*, 49, 49–75.
- Pástor, L. and Vorsatz, M. B. (2020), "Mutual fund performance and flows during the COVID-19 crisis," *The Review of Asset Pricing Studies*, 10, 791–833.
- Pástor, L., Stambaugh, R. F., and Taylor, L. A. (2021), "Sustainable investing in equilibrium," *Journal of Financial Economics*, 142, 550–571.
- Pástor, L., Stambaugh, R. F., and Taylor, L. A. (2022), "Dissecting green returns," *Journal of Financial Economics*, 146, 403–424.
- Pedersen, L. H., Fitzgibbons, S., and Pomorski, L. (2021), "Responsible investing: The ESG-efficient frontier," *Journal of Financial Economics*, 142, 572–597.
- Ramelli, S. and Wagner, A. F. (2020), "Feverish stock price reactions to COVID-19," *The Review of Corporate Finance Studies*, 9, 622–655.
- Renneboog, L., Ter Horst, J., and Zhang, C. (2011), "Is ethical money financially smart? Nonfinancial attributes and money flows of socially responsible investment funds," *Journal of Financial Intermediation*, 20, 562–588.

- Riedl, A. and Smeets, P. (2017), "Why do investors hold socially responsible mutual funds?" *The Journal of Finance*, 72, 2505–2550.
- Sautner, Z., van Lent, L., Vilkov, G., and Zhang, R. (2022a), "Firm-level Climate Change Exposure," *Journal of Finance, forthcoming*.
- Sautner, Z., Van Lent, L., Vilkov, G., and Zhang, R. (2022b), "Pricing climate change exposure," *Management science, forthcoming*.
- Seltzer, L., Starks, L. T., and Zhu, Q. (2021), "Climate regulatory risks and corporate bonds," *Working paper, University of Texas.*
- Stellner, C., Klein, C., and Zwergel, B. (2015), "Corporate social responsibility and Eurozone corporate bonds: The moderating role of country sustainability," *Journal of Banking & Finance*, 59, 538–549.
- van der Lugt, C., van de Wijs, P. P., and Petrovics, D. (2020), "Sustainability Reporting Policy: Global Trends in Disclosure as the ESG Agenda Goes Mainstream," *Carrots & Sticks*.
- Visco, I. (2019), "Sustainable development and climate risks: the role of central banks," Speech at 18th International Conference for Credit Risk Evaluation "Assessing and Managing Climate Change Risk: Opportunities for Financial Institutions".
- Zaghini, A. (2016), "Fragmentation and heterogeneity in the euro-area corporate bond market: Back to normal?" *Journal of Financial Stability*, 23, 51–61.
- Zaghini, A. (2021), "The Covid pandemic in the market: infected, immune and cured bonds," *ECB Working Paper*.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1369 Public guarantees and credit additionality during the Covid-19 pandemic, by Giuseppe Cascarino, Raffaele Gallo, Francesco Palazzo and Enrico Sette (April 2022).
- N.1370 The effects of local demand and supply restrictions on markup, by Antonio Acconcia and Elisa Scarinzi (June 2022).
- N. 1371 Mutual fund trading and ESG stock resilience during the Covid-19 stock market crash, by Rui Albuquerque, Yrjö Koskinen and Raffaele Santioni (June 2022).
- N. 1372 *Higher capital requirements and credit supply: evidence from Italy*, by Maddalena Galardo and Valerio Vacca (June 2022).
- N. 1373 Voluntary support and ring-fencing in cross-border banks, by Gyoengyi Loranth, Anatoli Segura and Jing Zeng (June 2022).
- N. 1374 PIt ain't where you're from it's where you're at: firm effects, state dependence, and the gender wage gap, by Sabrina Di Addario, Patrick Kline, Raffaele Saggio and Mikkel Søelvsten (June 2022).
- N. 1375 New facts on consumer price rigidity in the Euro Area, by Erwan Gautier, Cristina Conflitti, Riemer P. Faber, Brian Fabo, Ludmila Fadejeva, Valentin Jouvanceau, Jan-Oliver Menz, Teresa Messner, Pavlos Petroulas, Pau Roldan-Blanco, Fabio Rumler, Sergio Santork, Elisabeth Wieland and Hélène Zimmer (July 2022).
- N.1376 Connecting to power: political connections, innovation, and firm dynamics, by Ufuk Akcigit, Salomé Baslandze and Francesca Lotti (July 2022).
- N. 1377 "Green" fiscal policy measures and non-standard monetary policy in the Euro Area, by Anna Bartocci, Alessandro Notarpietro and Massimiliano Pisani (July 2022).
- N.1378 *Firm liquidity and the transmission of monetary policy*, by Margherita Bottero and Stefano Schiaffi (July 2022).
- N. 1379 *The impact of Covid-19 on the european short-term rental market*, by Elisa Guglielminetti, Michele Loberto and Alessandro Mistretta (July 2022).
- N.1380 An analysis of objective inflation expectations and inflation risk premia, by Sara Cecchetti, Adriana Grasso and Marcello Pericoli (July 2022).
- N. 1381 The role of central bank communication in inflation-targeting Eastern European emerging economies, by Valerio Astuti, Alessio Ciarlone and Alberto Coco (October 2022).
- N. 1382 The heterogeneous effects of bank mergers and acquisitions on credit to firms: evidence from Italian macro-regions, by Silvia Del Prete, Cristina Demma, Iconio Garrì, Marco Piazza and Giovanni Soggia (October 2022).
- N. 1383 Bank lending to small firms: metamorphosis of a financing model, by Paolo Finaldi Russo, Valentina Nigro and Sabrina Pastorelli (October 2022).
- N. 1384 *MMacroeconomic effects of growth-enhancing measures in the euro area*, by Alessandro Cantelmo, Alessandro Notarpietro and Massimiliano Pisani (October 2022).
- N.1385 *Revisiting the real exchange rate misalignment-economic growth nexus via the across-sector misallocation channel*, by Claire Giordano (October 2022).
- N. 1386 Labor market spillovers of a large plant opening. Evidence from the oil industry, by Matteo Alpino, Irene Di Marzio, Maurizio Lozzi and Vincenzo Mariani (October 2022).
- N. 1387 Unburdening regulation: the impact of regulatory simplification on photovoltaic adoption in Italy, by Federica Daniele, Alessandra Pasquini, Stefano Clò and Enza Maltese (October 2022).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2020

- ALESSANDRI P. and M. BOTTERO, *Bank lending in uncertain times,* R European Economic Review, V. 128, WP 1109 (April 2017).
- ANTUNES A. and V. ERCOLANI, Public debt expansions and the dynamics of the household borrowing constraint, Review of Economic Dynamics, v. 37, pp. 1-32, WP 1268 (March 2020).
- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities,* Journal of Business & Economic Statistics, v. 38, 4, pp. 826-838, **WP 974 (October 2014).**
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, Discretion and supplier selection in public procurement, Journal of Law, Economics, and Organization, v. 37, 1, pp. 134-166, WP 1178 (June 2018)
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, v. 16, 6, pp. 137-184, **WP 1201 (November 2018).**
- BOTTERO M., F. MEZZANOTTI and S. LENZU, Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy, Journal of International Economics, v. 126, article 103328, WP 1032 (October 2015).
- BRIPI F., D. LOSCHIAVO and D. REVELLI, Services trade and credit frictions: evidence with matched bank *firm data*, The World Economy, v. 43, 5, pp. 1216-1252, **WP 1110 (April 2017).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, *Venture capitalists at work: a Diff-in-Diff approach at late-stages of the screening process*, Journal of Business Venturing, v. 35, 3, **WP 1131 (September 2017).**
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome,* Journal of Regional Science, v. 60, 4, pp. 801-822, WP 1208 (February 2019).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis,* Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, **WP 1179 (June 2018).**
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, v. 102, **WP 1152 (December 2017).**
- D'ALESSIO G., *Measurement errors in survey data and the estimation of poverty and inequality indices,* Statistica Applicata - Italian Journal of Applied Statistics, v. 32, 3, **WP 1116 (June 2017).**
- DE BLASIO G., I. DE ANGELIS and L. RIZZICA, *Lost in corruption. Evidence from EU funding to Southern Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 6, 3, pp. 355–377, WP 1180 (December 2016).
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises*, Review of World Economic, v. 156, 4, pp. 859 885, WP 1217 (April 2019).
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- ERCOLANI V. and F. NATOLI, *Forecasting US recessions: the role of economic uncertainty*, Economics Letters, v. 193, WP 1299 (October 2020).
- MAKINEN T., L. SARNO and G. ZINNA, *Risky bank guarantees*, Journal of Financial Economics, v. 136, 2, pp. 490-522, **WP 1232 (July 2019).**
- MODENA F., E. RETTORE and G. M. TANZI, *The effect of grants on university dropout rates: evidence from the Italian case*, Journal of Human Capital, v. 14, 3, pp. 343-370, WP 1193 (September 2018).
- NISPI LANDI V., *Capital controls spillovers*, Journal of International Money and Finance, v. 109, WP 1184 (July 2018).
- PERICOLI M., On risk factors of the stock-bond correlation, International Finance, v. 23, 3, pp. 392-416, WP 1198 (November 2018).
- PIETRUNTI M. and F. M. SIGNORETTI, Unconventional monetary policy and household debt: the role of cashflow effects, Journal of Macroeconomics, v. 64, Article 103201, WP 1142 (October 2017).
- RAINONE E., *The network nature of OTC interest rates*, Journal of Financial Markets, v.47, article 100525, WP 1022 (July 2015).
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, **WP 1059 (March 2016).**
- RIZZICA L., *Raising aspirations and higher education. Evidence from the UK's widening participation policy,* Journal of Labor Economics, v. 38, 1, pp. 183-214, **WP 1188 (September 2018).**
- RIZZICA L. and M. TONELLO, Persuadable perceptions: the effect of media content on beliefs about

"TEMI" LATER PUBLISHED ELSEWHERE

corruption, Economic Policy, v. 35, 104, pp. 679-737, WP 1043 (November 2016).

- SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, v. 24, 4, pp. 773-811, WP 1146 (October 2017).
- SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, *Bank Quality, judicial efficiency and loan repayment delays in Italy,* Journal of Finance, v. 75, 4, pp. 2139-2178, **WP 1072 (July 2016).**

2021

- ACCETTURO A., A. LAMORGESE, S. MOCETTI and D. PELLEGRINO, *Housing Price elasticity and growth: evidence from Italian cities,* Journal of Economic Geography, v. 21, 3, pp. 367-396, WP 1267 (March 2020).
- AFFINITO M. and M. PIAZZA, Always look on the bright side? Central counterparties and interbank markets during the financial crisis, International Journal of Central Banking, v. 17, 1, pp. 231-283, WP 1181 (July 2018).
- ALBANESE G., E. CIANI and G. DE BLASIO, *Anything new in town? The local effects of urban regeneration policies in Italy*, Regional Science and Urban Economics, v. 86, **WP 1214 (April 2019)**.
- ALBANESE G., G. DE BLASIO and A. LOCATELLI, *Does EU regional policy promote local TFP growth? Evidence from the Italian Mezzogiorno*, Papers in Regional Science, v. 100, 2, pp. 327-348, **WP 1253 (December 2019).**
- ALBERTAZZI A., A. NOBILI and F. M. SIGNORETTI, *The bank lending channel of conventional and unconventional monetary policy*, Journal of Money, Credit and Banking, v. 53, 2-3, pp. 261-299, **WP 1094 (Jan 2017)**.
- ANZUINI A. and L. ROSSI, Fiscal policy in the US: a new measure of uncertainty and its effects on the American economy, Empirical Economics, v. 61, 6, pp. 2613-2634, WP 1197 (November 2018).
- APRIGLIANO V. and D. LIBERATI, Using credit variables to date business cycle and to estimate the probabilities of recession in real time, The Manchester School, v. 89, 51, pp. 76-96, WP 1229 (July 2019).
- AUER S., M. BERNARDINI and M. CECIONI, *Corporate leverage and monetary policy effectiveness in the euro area,* European Economic Review, v. 140, Article 103943, **WP 1258 (December 2019).**
- BANERJEE R, L. GAMBACORTA and E. SETTE, *The real effects of relationship lending*, Journal of Financial Intermediation, v. 48, Article 100923, WP 1133 (September 2017).
- BARONE G., F. DAVID, G. DE BLASIO and S. MOCETTI, *How do house prices respond to mortgage supply?*, Journal of Economic Geography, v. 21, 1, pp.127-140, **WP 1282 (June 2020).**
- BARONE G. and S. MOCETTI, Intergenerational mobility in the very long run: Florence 1427-2011, Review of Economic Studies, v. 88, 4, pp. 1863–1891, WP 1060 (April 2016).BARTOCCI A., L. BURLON, A. NOTARPIETRO and M. PISANI, Macroeconomic effects of non-standard monetary policy measures in the Euro Area: the role of corporate bond purchases, The Manchester School, v. 89, S1, pp. 97-130, WP 1241 (Oct 2019).
- BATINI N., A. CANTELMO, G. MELINA and S. VILLA, *How loose, how tight? A measure of monetary and fiscal stance for the euro area,* Oxford Economic Papers, v. 73, 4, pp. 1536-1556, **WP 1295 (September 2020).**
- BENETTON M. and D. FANTINO, *Targeted monetary policy and bank lending behavior*, Journal of Financial Economics, v. 142, 1, pp. 404-429, **WP 1187 (September 2018).**
- BUSETTI F., M. CAIVANO and D. DELLE MONACHE, Domestic and global determinants of inflation: evidence from expectile regression, Oxford Bulletin of Economics and Statistics, v. 83, 4, pp. 982-1001, WP 1225 (June 2019).
- BUSETTI F., M. CAIVANO, D. DELLE MONACHE and C. PACELLA, *The time-varying risk of Italian GDP*, Economic Modelling, v. 101, Article 105522, WP 1288 (July 2020).
- BUSETTI F., S. NERI, A. NOTARPIETRO and M. PISANI, Monetary Policy strategies in the new normal: a modelbased analysis for the Euro Area, Journal of Macroeconomics, v. 70, Article 103366, WP 1308 (December 2020).
- CAPOLONGO A. and C. PACELLA, Forecasting inflation in the Euro Area: countries matter, Empirical Economics, v. 61, 4, pp. 2477-2499, WP 1224 (June 2019).
- CARMIGNANI A., G. DE BLASIO, C. DEMMA and A. D'IGNAZIO, *Urbanization and firm access to credit,* Journal of Regional Science, v. 61, 3, pp. 597-622, WP 1222 (June 2019).
- CORNELI F., *Financial integration without financial development*, Atlantic Economic Journal, v. 49, 2, pp. 201-220, **WP 1120 (June 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, Secular stagnation, R&D, public investment and monetary policy: a global-model perspective, Macroeconomic Dynamics, v. 25, 5, pp. 1267-1287, WP 1156 (December 2017).

- DE PHILIPPIS M., *Multitask agents and incentives: the case of teaching and research for university professors,* Economic Journal, v. 131, 636, pp. 1643-1681, **WP 1042 (December 2015).**
- DEL PRETE S. and M. L. STEFANI, Women as "Gold Dust": gender diversity in top boards and the performance of Italian banks, Economic Notes, Monte dei Paschi di Siena, v. 50, 2, e12183, WP 1014 (June 2015).
- FERRERO G., M. LOBERTO and M. MICCOLI, *The assets' pledgeability channel of unconventional monetary policy*, Economic Inquiry, v. 59, 4, pp. 1547-1568, **WP 1119 (June 2017).**
- FIDORA M., C. GIORDANO and M. SCHMITZ, *Real exchange rate misalignments in the Euro Area,* Open Economies Review, v. 32, 1, pp. 71-107, **WP 1162 (January 2018).**
- GAMBACORTA L., G. RICOTTI, S. SUNDARESAN and Z. WANG, *Tax effects on bank liability structure*, European Economic Review, v. 138, Article 103820, **WP 1101 (February 2017).**
- HERTWECK M., V. LEWIS and S. VILLA, *Going the extra mile: effort by workers and job-seekers,* Journal of Money, Credit and Banking, v. 54, 8, pp. 2099-2127, **WP 1277 (June 2020).**
- LI F., A. MERCATANTI, T. MAKINEN and A. SILVESTRINI, *A regression discontinuity design for ordinal running variables: evaluating central bank purchases of corporate bonds,* The Annals of Applied Statistics, v. 15, 1, pp. 304-322, **WP 1213 (March 2019).**
- LOSCHIAVO D., *Big-city life (dis)satisfaction? The effect of urban living on subjective well-being*, Journal of Economic Behavior & Organization, vol. 192, pp. 740-764, **WP 1221 (June 2019).**
- LOSCHIAVO D., *Household debt and income inequality: evidence from Italian survey data*, Review of Income and Wealth. v. 67, 1, pp. 61-103, **WP 1095 (January 2017).**
- METELLI L. and F. NATOLI, *The international transmission of US tax shocks: a proxy-SVAR approach*, IMF Economic Review, v. 69, 2, pp. 325-356, **WP 1223 (June 2019).**
- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, v. 32, 1, pp. 183-211, WP 1200 (November 2018).
- PAPETTI A., *Demographics and the natural real interest rate: historical and projected paths for the Euro Area*, Journal of Economic Dynamics and Control, v. 132, Article 04209, WP 1306 (November 2020).
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, v. 39, 2, pp. 575-588, WP 1092 (January 2017).

2022

- ANDINI M., M. BOLDRINI, E. CIANI, G. DE BLASIO, A. D'IGNAZIO and A. PALADINI, Machine learning in the service of policy targeting: the case of public credit guarantees, Journal of Economic Behavior & Organization, v. 198, pp. 434-475, WP 1206 (February 2019).
- ANGELICO C., J. MARCUCCI, M. MICCOLI and F. QUARTA, Can we measure inflation expectations using twitter?, Journal of Econometrics, v. 228, 2, pp. 259-277, WP 1318 (February 2021).
- BOTTERO M., C. MINOIU, J. PEYDRÒ, A. POLO, A. PRESBITERO and E. SETTE, *Expansionary yet different: credit supply and real effects of negative interest rate policy,* Journal of Financial Economics, v. 146, 2, pp. 754-778, **WP 1269 (March 2020).**
- CANTELMO A., *Rare disasters, the natural interest rate and monetary policy,* Oxford Bulletin of Economics and Statistics, v. 84, 3, pp. 473-496, **WP 1309 (December 2020).**
- CARRIERO A., F. CORSELLO and M. MARCELLINO, *The global component of inflation volatility*, Journal of Applied Econometrics, v. 37, 4, pp. 700-721, **WP 1170 (May 2018).**
- CIAPANNA E. and G. ROVIGATTI, *The grocery trolley race in times of Covid-19. Evidence from Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 8, 2, pp. 471-498, **WP 1341 (June 2021)**.
- FAIELLA I. and A. MISTRETTA, The net zero challenge for firms' competitiveness, Environmental & Resource Economics, v. 83, pp. 85-113, WP 1259 (February 2020).
- GUISO L., A. POZZI, A. TSOY, L. GAMBACORTA and P. E. MISTRULLI, *The cost of steering in financial markets:* evidence from the mortgage market, Journal of Financial Economics, v.143, 3, pp. 1209-1226, **WP 1252 (December 2019).**
- LAMORGESE A. and D. PELLEGRINO, *Loss aversion in housing appraisal: evidence from Italian homeowners,* Journal of Housing Economics, v. 56, Article 101826, WP 1248 (November 2019).
- LI F., T. MÄKINEN, A. MERCATANTI and A. SILVESTRINI, *Causal analysis of central bank holdings of corporate bonds under interference,* Economic Modelling, v.113, Article 105873, WP 1300 (November 2020).
- MIRENDA L., M. SAURO and L. RIZZICA, *The economic effects of mafia: firm level evidence*, American Economic Review, vol. 112, 8, pp. 2748-2773, **WP 1235 (October 2019).**
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility,* Journal of Human Resources, v. 57, 2, pp. 525-554, WP 1182 (July 2018).

- PERICOLI M. and M. TABOGA, *Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models*, Journal of Financial Econometrics, v. 20, 5, pp. 807-838, WP 1189 (September 2018).
- ROSSI P. and D. SCALISE, *Financial development and growth in European regions*, Journal of Regional Science, v. 62, 2, pp. 389-411, WP 1246 (November 2019).
- SCHIVARDI F., E. SETTE and G. TABELLINI, *Credit misallocation during the European financial crisis,* Economic Journal, v. 132, 641, pp. 391-423, **WP 1139 (September 2017).**
- TABOGA M., Cross-country differences in the size of venture capital financing rounds: a machine learning approach, Empirical Economics, v. 62, 3, pp. 991-1012, WP 1243 (November 2019).

FORTHCOMING

- APRIGLIANO V., S. EMILIOZZI, G. GUAITOLI, A. LUCIANI, J. MARCUCCI and L. MONTEFORTE, *The power of textbased indicators in forecasting Italian economic activity*, International Journal of Forecasting, WP 1321 (March 2021).
- BRONZINI R., A. D'IGNAZIO and D. REVELLI, *Financial structure and bank relationships of Italian multinational firms*, Journal of Multinational Financial Management, **WP 1326 (March 2021).**
- DI ADDARIO S., P. KLINE, R. SAGGIO and M. SØLVSTEN, *It ain't where you're from, it's where you're at: hiring origins, firm heterogeneity, and Wages, Journal of Econometrics,* WP 1374 (June 2022).
- FERRARI A. and V. NISPI LANDI, *Toward a green economy: the role of central bank's asset purchases,* International Journal of Central Banking, **WP 1358 (February 2022).**
- LILLA F., Volatility bursts: a discrete-time option model with multiple volatility components, Journal of Financial Econometrics, WP 1336 (June 2021).
- LOBERTO M, Foreclosures and house prices, Italian Economic Journal / Rivista italiana degli economisti, WP 1325 (March 2021).
- LOBERTO M, A. LUCIANI and M. PANGALLO, *What do online listings tell us about the housing market?*, International Journal of Central Banking, **WP 1171 (April 2018).**
- RIZZICA L., G. ROMA and G. ROVIGATTI, *The effects of shop opening hours deregulation: evidence from Italy*, The Journal of Law and Economics, **WP 1281 (June 2020).**
- TANZI G. M., Scars of youth non-employment and labour market conditions, Italian Economic Journal / Rivista italiana degli economisti, WP 1312 (December 2020).